

JGR Biogeosciences

RESEARCH ARTICLE

10.1029/2019JG005298

Key Points:

- Recent weakening of the relationship between climatic water balance and radial growth and C isotopes was species specific
- Water availability is the dominant factor influencing long-term radial growth even when considering the simultaneous influence of other factors such as increasing atmospheric CO₂ and decreasing acidic deposition
- Increasing atmospheric CO₂ concentrations increased water-use efficiency but not growth of most tree species

Supporting Information:

• Supporting Information S1

Correspondence to:

J. T. Maxwell, maxweljt@indiana.edu

Citation:

Maxwell, J. T., Harley, G. L., Mandra, T. E., Yi, K., Kannenberg, S. A., Au, T. F., et al (2019). Higher CO₂ Concentrations and Lower Acidic Deposition Have Not Changed Drought Response in Tree Growth But Do Influence iWUE in Hardwood Trees in the Midwestern United States. *Journal of Geophysical Research: Biogeosciences*, *124*, 3798–3813. https://doi.org/10.1029/2019JG005298

Received 5 JUN 2019 Accepted 31 OCT 2019 Accepted article online 6 NOV 2019 Published online 13 DEC 2019

Higher CO₂ Concentrations and Lower Acidic Deposition Have Not Changed Drought Response in Tree Growth But Do Influence iWUE in Hardwood Trees in the Midwestern United States

Justin T. Maxwell^{1,2}, Grant L. Harley³, Tessa E. Mandra², Koong Yi⁴, Steven A. Kannenberg⁵, Tsun Fung Au¹, Scott M. Robeson¹, Neil Pederson², Peter E. Sauer⁶, and Kimberly A. Novick⁷

¹Department of Geography, Indiana University, Bloomington, IN, USA, ²Harvard Forest, Harvard University, Petersham, MA, USA, ³Department of Geography, The University of Idaho, Moscow, ID, USA, ⁴Department of Environmental Sciences, University of Virginia, Charlottesville, VA, USA, ⁵School of Biological Sciences, University of Utah, Salt Lake City, UT, USA, ⁶Department of Earth and Atmospheric Sciences, Indiana University, Bloomington, IN, USA, ⁷School of Public and Environmental Affairs, Bloomington, IN, USA

Abstract Several important environmental influences of tree growth and carbon sequestration have changed over the past several decades in eastern North America, specifically, more frequent pluvial conditions, increased carbon dioxide (CO₂) concentrations, and decreased acidic deposition. These factors could lead to changes in the relationship between tree growth and water availability, and perhaps even decouple the two, having large implications on how future climate change will impact forest productivity and carbon sequestration. Here, we examine the concurrent influence of the climatic water balance (precipitation minus potential evapotranspiration), CO₂ concentrations, and sulfate and nitrogen deposition on radial tree growth, carbon isotopes, and intrinsic water-use efficiency (iWUE) for several hardwood tree species in the Midwestern United States. We found that when considering the simultaneous influence of these factors, the climatic water balance is the dominant influence on annual growth. Therefore, the recent pluvial period is the primary cause of the weakening relationship between radial growth and water availability. Even during pluvial periods, water availability is the primary control on growth, with increasing CO_2 concentrations and decreased SO_4 deposition being secondary factors. Importantly, the weakening in the climate-growth relationship is species specific, with Acer species having stable relationships with the climatic water balance, Liriodendron tulipifera showing a strengthening relationship, and Quercus species and Populus grandidentata exhibiting weakening. Thus, interannual variations in soil moisture unevenly impact tree growth and carbon sequestration. Our findings suggest that, despite recent pluvial conditions, increasing CO₂ concentrations and decreasing acidic deposition have not buffered the impact of water availability on tree growth and carbon sequestration.

Plain Language Summary In recent decades, many factors that influence tree growth have changed across the Midwestern United States, including more precipitation, higher carbon dioxide concentrations in the atmosphere, and fewer pollutants. Changes in these environmental factors could result in tree growth being less sensitive to water availability. While a weakening relationship between water availability and tree growth is present, it is difficult to determine the cause. Here, we examine the simultaneous influence of a wetter climate, higher carbon dioxide concentrations, and a decrease in pollutants deposition on tree growth and how efficiently trees use water. We found that when considering all three variables, increased moisture was the leading influence on tree growth. Therefore, the recent wet period is behind trees being less sensitive to soil moisture, not increases in carbon dioxide or decreases in pollutants. It is important to note that some species such as oak and aspen did become less sensitive to soil moisture, other species such as maple and tuliptree did not. This is important because when drought conditions return to the region, trees will still be sensitive despite higher carbon dioxide concentrations or decreases in sulfate and nitrate deposition.

©2019. American Geophysical Union. All Rights Reserved.



1. Introduction

Tree rings in mesic, temperate forests are a useful tool for estimating past climatic conditions, including water availability (Cook, 1991; Cook & Jacoby, 1977; Maxwell et al., 2011; Pederson et al., 2013; Stahle & Cleaveland, 1992). To accommodate and survive extreme hydroclimatic variability, trees alter their stomatal conductance and photosynthetic assimilation rates, resulting in changes in primary and secondary tree growth (Bréda et al., 2006) and the stable isotopic composition of annual tree rings (Belmecheri et al., 2014; Gagen et al., 2011; Levesque et al., 2019; Lévesque et al., 2014; Loader et al., 2007; Yi et al., 2019). Therefore, radial growth and stable isotopes within tree rings can provide long-term information on the influence of seasonal and interannual changes in climate and environmental conditions on forest productivity. Water availability is an important driver of forest productivity and carbon sequestration even in the mesic Midwestern United States (hereafter Midwest) and is likely to become more influential in the future due to ongoing increases in air temperature and associated increases in vapor pressure deficit (VPD) and potential evapotranspiration (Allen et al., 2015; Choat et al., 2012; Choat et al., 2018; Clark et al., 2016; McDowell & Allen, 2015; Novick et al., 2016). In the Midwestern United States, radial growth from multiple species has been shown to be responsive to water availability (Maxwell, 2016; Maxwell et al., 2015). However, the last several decades have been the wettest in recorded history in the Midwest (Andersen et al., 2012; Karl et al., 1996; Mishra et al., 2010) and potentially the wettest in the last several centuries (Ford, 2014; Maxwell & Harley, 2017; McEwan et al., 2011). The recent pluvial period could be causing tree growth across the Midwest to appear less sensitive to soil moisture (Maxwell et al., 2016).

However, many other factors that influence tree growth have changed over the last several decades in the Midwest, making the attribution of a weakening relationship difficult. Pollution such as acidic deposition can negatively influence tree growth and the ability of trees to tolerate climatic stress (Cook et al., 1987; Johnson et al., 1988; LeBlanc et al., 1987). Acidic deposition has decreased in recent decades in the Midwest and this could have resulted in an increase in tree growth (Engel et al., 2016; Kosiba et al., 2018; Mathias & Thomas, 2018; Thomas et al., 2013), although other studies have shown little to no impact on growth (Bishop et al., 2015; Schaberg et al., 2014). Further, nitrogen deposition can act as a fertilizer to increase carbon storage of trees (Horn et al., 2018; Thomas et al., 2010). However, the ability of nitrogen deposition to increase growth appears modest (Caspersen et al., 2000; Hyvönen et al., 2008; Ollinger et al., 2002; Walker et al., 2019). Increasing atmospheric CO₂ concentrations may also increase tree growth (Ainsworth & Long, 2005; Graumlich, 1991; Soulé & Knapp, 2006), however, the relationship is not spatially uniform (Peñuelas et al., 2011) and complexities arise when multiple factors interact to influence growth (McCarthy et al., 2010; Norby & Zak, 2011; Zhu et al., 2016). When examining the simultaneous impacts of climate variability, CO₂ concentrations, and pollutants, Levesque et al. (2017) found the primary driver of tree growth at one site in the Northeast United States to be water availability. However, it is unclear if water availability is driving the documented weakening relationship between tree rings and soil moisture in the Midwest (Maxwell et al., 2016).

Moreover, the response of trees to these multiple drivers may vary from one species to the next. It is increasingly clear that key species growing in eastern United States forests adopt different approaches to respond to drought. For example, oaks (*Quercus*)—a keystone ecological species—tend to adopt an "anisohydric" drought response strategy that is characterized by relatively high rates of stomatal conductance and photosynthesis, at the expense of excessive water loss and potential damage to the tree's hydraulic system (Meinzer et al., 2013; Thomas et al., 2013; Brzostek et al., 2014; Roman et al., 2015). On the other hand, more "isohydric" species such as tulip poplar (*Liriodendron tulipifera*) close stomates quickly during drought stress, thereby limiting carbon uptake (Yi et al., 2019). However, we still have much to learn about how these leaf-level strategies translate into long-term growth responses (Kannenberg et al., 2019) and know very little about the interactions between drought response strategies and variation in atmospheric CO₂ or nitrogen deposition.

The recent pluvial period has been identified as a potential cause of the weakening relationship between tree rings and soil moisture (Maxwell et al., 2016). However, changes in CO_2 concentrations and nitrogen and sulfate deposition in recent decades could also independently enhance tree growth. In this paper, we examine the concurrent impacts of changes in moisture availability, CO_2 concentrations, and nitrogen and sulfate





Figure 1. Map of study site location (left), where stars represent sites. For each site, a time series of sulfate (SO_4) deposition (top right), nitrogen (N) deposition (second right), climatic water balance (P-PET) (third right) is shown, along with the global average CO_2 concentrations (bottom right).

deposition on tree growth and stable carbon isotopic composition across three sites in the Midwest. We target important canopy-dominant species across the Midwest to test the hypothesis that water availability is the driving factor influencing radial growth when considering the simultaneous influence of other variables. We also test a secondary hypothesis: Trees that employ a more conservative isohydric water-use strategy will be more sensitive to water availability than those that use a more anisohydric strategy.

2. Data and Methods

2.1. Study Sites

We studied a variety of broadleaf tree species at three sites across the central hardwood region in the Midwest (Figure 1, left). Each site exists as a secondary to tertiary forest (i.e., forests that have regrown after timber harvest) and includes an Ameriflux tower: the Missouri Ozark Site (US-MOz), the Morgan-Monroe State Forest (US-MMS), and the University of Michigan Biological Station (US-UMB). Many tree species occur at each site, but we focused our sampling efforts to the two to three most common species with the criteria of having a relatively anisohydric species (i.e., *Quercus* species) and more isohydric codominants (i.e., *Acer* species, *Liriodendron tulipifera*). We sampled *Acer* saccharum and *Quercus* alba at MOz; *A.* saccharum, *Q.* alba, and *Liriodendron tulipifera* at MMS; and *A.* rubrum, *Q.* rubra, and *Populus* grandidentata at UMB. Generally, there was a gradient of drought occurrence across the sites. MOz experienced the most severe drought conditions, MMS experienced intermediate drought, and UMB was absent of recent drought, while all sites experienced similar CO₂ concentrations and nitrogen and sulfate deposition (Figure 1).

2.2. Dendrochronological Analyses

At each study site, we extracted cores from 5–15 trees of each species using a 4.3 mm diameter handheld increment borer. We targeted healthy canopy-dominant trees within each Ameriflux tower footprint and extracted two cores on opposite sides, parallel to the slope to help reduce the effect of asymmetric growth. Although sampling biases from only selecting canopy-dominant trees can arise, these are typically minimized when examining growth responses to environmental variability (Nehrbass-Ahles et al., 2014). Tree cores were air dried, glued to wooden mounts, and sanded using progressively finer sandpaper until the ring structure was visible under 10X magnification. For each species, we visually crossdated the growth rings from each sample core using the list method (Yamaguchi, 1991). We then statistically confirmed the



5 5					
		<i>n</i> of trees (cores)		Mean	Interseries
Site name	Species	Ring width	Isotopes	age	correlation
Missouri Ozarks (MOz)	A. saccharum	8(15)	4	102	0.50
	Q. alba	10(20)	4	92	0.68
Morgan-Monroe State Forest (MMS)	A. saccharum	9(17)	4	87	0.49
	Q. alba	12(21)	4	69	0.58
	L. tulipifera	10(20)	4	65	0.62
University of Michigan Biological Station (UMB)	A. rubrum	5(9)	4	81	0.64
	Q. rubra	5(9)	4	88	0.56
	P. grandidentata	6(10)	4	71	0.45

Table 1

Summary of Tree Characteristics and Crossdating Statistics

visual crossdating using the program COFECHA (Holmes, 1983), which calculates correlation coefficients of ring widths between a given sample and the remaining samples from different trees. To remove the agerelated growth trend and the nonclimatic growth related to forest dynamics, we used a two thirds length spline with a frequency cutoff point of 50% (Cook, 1981) using the "dplR" package (Bunn, 2008) in R (R Core Team, 2017). We used the standardized chronology as it has been shown to retain the strongest climate signal in the Midwest, including this study region (Cook, 1985; Maxwell et al., 2015).

To ensure that the variance in growth is not influenced by sample size, we restricted the analysis of tree rings to the period where are at least five trees (10 sample cores) are represented. To determine to what degree this standardization procedure could impact results, we also converted ring width to basal area increment (BAI) and removed only age-related trends using a Regional Curve Standardization approach (Peters et al., 2015). For each species, we calculated the regional curve, which is the average ontogenetic growth curve, by assigning the raw BAI measurements of each tree to the appropriate biological age. Each individual raw BAI series was then divided by the average regional curve to get the Regional Curve Standardization residual BAI series and then averaged across samples to get the standardized BAI chronology for each species (Briffa et al., 1992; Cook & Briffa, 1990).

2.3. Isotopic Analyses

We used tree ring δ^{13} C to calculate the carbon isotope discrimination (Δ^{13} C) from the latewood of each annual ring from 1970–2015. We used only the latewood portion of the annual tree ring because this portion is more responsive to environmental factors (Battipaglia et al., 2009; Belmecheri et al., 2014; Lipp et al., 1991; Livingston & Spittlehouse, 1996). For the species with ring porous wood type (e.g., *Q. alba*), we sliced the latewood portion of the growth ring from each year (e.g., post vessel element rows) and for the diffuse porous species, we sliced the last third of the growth ring due to the lack of a distinction between earlywood and latewood (cf.Levesque et al., 2017; Levesque et al., 2019; Yi et al., 2019). We combined the latewood samples from the two radial cores for each tree for δ^{13} C analysis. For each tree (n = 4; Table 1), α -cellulose was extracted using a modified Soxhlet method (Loader et al., 1997). Using an elemental analyzer (ECS4010; Costech, Valencia, CA, USA) coupled to an isotope-ratio mass spectrometer (Finnigan DELTAplusXP; ThermoFisher Scientific, Waltham, MA, USA), we measured the isotopic signature of the CO₂ produced by α -cellulose composition to obtain δ^{13} C from 1970–2015.

During photosynthesis, ¹²C atoms are used preferentially over ¹³C atoms and therefore calculating carbon isotope discrimination (Δ^{13} C) from tree rings allows us to look at the impacts of changes in photosynthetic carbon fixation over time. Δ^{13} C is calculated by taking the difference between the δ^{13} C of the air ($\delta^{13}C_{air}$) and δ^{13} C in tree ring ($\delta^{13}C_{tree}$). We obtain Δ^{13} C with the following equation:

$$\Delta^{13}C = \frac{\left(\delta^{13}C_{air} - \delta^{13}C_{tree}\right)}{\left(1 + \frac{\delta^{13}C_{tree}}{1000}\right)} \tag{1}$$

where $\delta^{13}C_{air}$ is the atmospheric $\delta^{13}C$ obtained from the Mauna Loa observatory (Keeling et al., 2005) and $\delta^{13}C_{tree}$ is the $\delta^{13}C$ from the α -cellulose tree ring samples. We estimated intercellular CO₂ concentration (*c*_i) by inverting the Farquhar et al. (1982) model of $\Delta^{13}C$ in C₃ plants:



$$\Delta^{13}C = a + (b-a)\frac{c_i}{c_a} \tag{2}$$

where *a* is a constant of 4.4‰ for the fractionation during CO_2 diffusion through the stomata (O'Leary 1981), *b* is a constant of 27‰ for the fractionation by ribulose 1,5-biphosphate carboxylase (Farquhar & Richards, 1984), and c_a are the ambient CO_2 concentrations. The intercellular CO_2 concentration was then used to calculate intrinsic water-use efficiency (iWUE), which is the ratio of net photosynthesis and conductance to water vapor:

$$iWUE = \frac{A}{g_s} = \frac{(c_a - c_i)}{1.6}$$
 (3)

where *A* is the rate of CO_2 assimilation and g_s is the rate of leaf stomatal conductance to water. As the ratio of photosynthesis to stomatal conductance, intrinsic water-use efficiency is a useful indicator of how droughtdriven declines in stomatal function (e.g., due to declining soil moisture) ultimately impact carbon uptake. Because the relationship between photosynthesis and conductance is nonlinear and tends to plateau when conductance is high, reduction to conductance (i.e., from unfavorable climatic water balance) should promote increases in iWUE while photosynthetic capacity itself is unaltered (Yi et al., 2019). At the same time, increase in photosynthesis that is not accompanied by reduction in stomatal conductance will also increase iWUE (Guerrieri et al., 2019). We show time series of the raw tree ring and isotopic data in supporting information Figure S1.

2.4. Climate and Deposition Data

We gathered mean monthly minimum and maximum temperature (Tmin and Tmax), maximum VPD, and total monthly precipitation (*P*) for the period 1895–2015. For each site, data for the nearest 4×4 km grid point from the PRISM data sets were used (http://prism.orgonstate.edu). Using the PRISM monthly mean minimum and maximum temperatures, we calculated potential evapotranspiration (PET) according to the Hargreaves formula (Hargreaves, 1994) using the "SPEI" package in R (Beguería et al., 2014; Vicente-Serrano et al., 2010). We then represented the climatic water balance as the difference between monthly precipitation and monthly potential evapotranspiration (i.e., P-PET). We gathered values of annual global average CO₂ concentrations (www.columbia.edu/~mhs119/GHGs/), and the water year (previous October to current September) total wet deposition of SO₄ and inorganic N from the National Atmospheric Deposition Program (Schwede & Lear, 2014) for the three 12 × 12 km grid points closest to our tree ring sampling sites.

2.5. Statistical Analyses

To examine the seasonal responses of the standardized growth metrics (i.e., standardized tree ring width and standardized BAI) and the isotopic metrics (Δ^{13} C and iWUE) to the monthly climate variables, we used Pearson's correlation analysis with bootstrapped estimates of significance using the R package "treeclim" (Zang & Biondi, 2015) for the common period between the climate data and the standardized growth and isotopic metrics (i.e., 1970-2015). The months of June, July, and August had the strongest correlations with the standardized growth and isotopic metrics and we, therefore, created a seasonal average over these months (hereafter: JJA). To examine potential differences in early and late portions of the growing season, we also created averages for May-June (MJ) and August-September (AS). All standardized growth and isotopic metrics and climate variables were prewhitened to remove the first-order autocorrelation before the correlation analysis. To determine if these species show a weakening relationship with water availability, we conducted a 40-year moving window correlation analysis between the standardized tree ring widths and BAI and the JJA P-PET for the period of overlap between the growth metrics and the climate data. Because all trees sampled were younger than the full length of the instrumental record, the common period was limited by the growth metrics and determined by only using growth data for the period were at least five trees were represented. However, using the longer growth metrics allows the assessment of a longer period due to the limited record of isotopic data.

Interpretation of moving correlation analysis can vary depending on the window size. Therefore, to evaluate the robustness of a potential change in the response of climate variability, we also used a state-space model with time varying parameter regression (Durbin and Koopman, 2001) and the Kalman filter (Kalman, 1960) to examine the time dependence of the relationship between tree growth and JJA P-PET using the R package





Figure 2. Hypothetical example of the structural equation model used to simultaneously assess the influence of climatic water balance, atmospheric CO_2 concentrations, SO_4 and N wet deposition on tree growth (standardized tree rings and basal area increment), and gas exchange inferred by $\Delta^{13}C$ and intrinsic water-use efficiency from tree rings. Double-headed black lines represent covariance relationship between response variables, and single-headed black lines indicate a causal relationship. Gray double-headed lines represent covariance between explanatory variables.

"dlm" (Petris, 2010). The time varying method with the Kalman filter has been successfully used to examine the temporal variability of climategrowth relationships of tree rings (Bishop et al., 2015; Cook & Johnson, 1989; Visser et al., 2010). The Kalman filter relaxes the least squares assumption that assumes that growth rate is consistently determined by the most limiting factor, allowing the detection of changes in the responses of radial growth to a particular climate variable. We applied the Kalman filter to the standardized tree ring widths and BAI chronologies and JJA P-PET for the overlapping period between growth and climate data. We found nearly identical results and therefore present only those for the tree ring widths.

To determine how simultaneous changes in P-PET, CO_2 concentration, and SO_4 and N deposition could be interacting to influence tree growth or isotopic composition, we performed piecewise structural equation modeling (SEM) using the R package "piecewiseSEM" (Lefcheck, 2016). Piecewise SEMs are ideal for small sample sizes. For each species at each site, we developed a piecewise SEM model (i.e., eight models) in order to assess the combined impacts of changing P-PET, atmospheric CO_2 concentrations, and SO_4 and N deposition on the standardized growth and

isotopic measurements (Figure 2). Because we have two growth metrics (tree ring width and BAI) and two isotopic metrics (Δ^{13} C and iWUE), we created piecewise SEMs for all possible combinations of the four metrics to determine if the results were sensitive to the metrics that were used. In any given SEM, we had one growth and one isotopic metric, resulting in a total of 32 models (four combinations of paired responses over eight species models). We used a linear mixed effects model (Zurr et al., 2009) to assess both the fixed effects (P-PET, CO₂, SO₄, and N) and random effects (individual trees) for each species at each site using the function *lme* in the "nlme" package (Pinheiro, 2013). All SEMs were fit for the period of overlap between all variables of 1985–2015. We used Fisher's C statistic and Akaike's information criterion (AIC) to assess model fit.

We also included Tmax as a variable in the SEMs to see if the inclusion would increase variance explained in the SEMs. We found that some of the models included Tmax as a possible explanatory variable, but in every one of these models Tmax replaced P-PET (i.e., T and P-PET were never both included in the same model). In the models that retained Tmax, we compared the AIC and R^2 of the two models and found that the AIC and R^2 indicated that the model with P-PET was a better fit. Thus, the Tmax response is more closely linked to increasing water demand and therefore we only included P-PET.

3. Results

3.1. Climate Analyses

All eight species chronologies statistically crossdated and produced interseries correlation values of 0.45–0.65, indicating that radial growth from individual trees for each species is responding similarly to environmental variability (Table 1). The months JJA had the highest correlation values with both the growth and isotopic metrics (Figure 3 and supporting information Figures S2–S10). However, the exception is with maximum temperatures, where the isotopic metrics had higher correlations in the earlier and later growing season (i.e., MJ and AS) compared to JJA (Figure 3). Generally, the early portion of the growing season (MJ) was more important to both growth and isotopic metrics than the later portion (AS), with the exceptions of maximum temperature and VPD. Interestingly, P and P-PET had stronger correlations with the growth metrics, while Tmax and VPD had a stronger influence on the isotopic metrics (Figure 3), potentially reflecting the importance of atmospheric water demand on stomatal conductance.

To evaluate the influence of water balance on both growth and isotopic metrics, we calculated the JJA average of the climatic water balance (P-PET). Moving correlations between the growth metrics and JJA P-PET suggested that the strength of the relationship between growth and climatic water balance for most species had become weaker with time (Figure 4). In contrast, *A. saccharum* at MOz had a consistent correlation through time as did *Q. alba* at MMS, albeit weak, while *L. tulipifera* at MMS had an increasing correlation through time (Figure 4). In addition to general patterns in the correlations, the differences in correlation







values varies across sites and species. Generally, the *Acer* genus had less variance with the exception of MMS ACSA (*Acer saccharum*) (Table 2). The Kalman filter approach confirmed the overall declining relationships found in the moving correlation analysis, with the only difference being *A. rubrum* at UMB displayed a significant and stable relationship over the entire common period (Figure 5).

When examining the moving correlations between the growth metrics and P-PET for the early (MJ) and late (AS) portions of the growing season, we found that most species were temporally stable in the early portion but species at MMS had larger variances (Figure 6). Moving correlations of the later season showed both siteand species-specific differences. Correlation values increased through time for both species at MOz, while all other species at UMB and MMS decreased, with the exception of MMS LITU (Figure 6).

3.2. SEM

When examining the potential factors influencing growth simultaneously in the piecewise SEMs, we found that P-PET was the dominant influence (i.e., it had the highest standardized coefficients) of both metrics of growth for the majority of species at MOz and MMS, except for MOz *Q. alba*, where CO₂ was slightly stronger than P-PET (Figure 7 and supporting information Figures S10–S12). The SEMs generally explained between 15% to 50% of variance for the species at MOz and MMS with *L. tulipifera* being the most sensitive. In that



Journal of Geophysical Research: Biogeosciences



Figure 4. Moving correlations of JJA P-PET with standardized tree rings (left column) and standardized basal area increment (right column). Correlations were done in 40-year moving windows with the year on the *x* axis representing the last year of the interval. The length of the time series was determined by the expressed population signal values of >0.85. ACSA = Acer saccharum; QUAL = Quercus alba; LITU = Liriodendron tulipifera; ACRU = Acer rubrum; QURU = Quercus rubra; POGR = Populus grandidentata.

case, SEMs explained >30–60% of the variance (Figure 7). However, for UMB, all the variables had a weak influence on growth with SEMs explaining <20% of the variance, with the exception of POGR BAI. Further, only one species at UMB (*P. grandidentata*) had a P-PET as a variable that significantly influenced any of the growth metrics, while SO₄ and CO₂ were significant variables for *P. grandidentata* and *Q. rubra*, respectively, and CO₂ was significantly related to tree ring width for *A. rubrum* (Figure 7 and supporting information Figures S10–S12). In general, climatic water balance, CO₂, and SO₄ were the most consistent variables retained in the SEMs in UMB although the variance explained was generally low.

The variables influencing iWUE were not consistent across species except CO_2 concentration, which was present in the majority of the SEMS (Figure 7 and supporting information Figures S10–S12). Moreover,

Summary of the Range of Correlation Values From the Moving Correlation Analysis						
Species	Metric	Maximum correlation	Minimum correlation	Difference (Max – Min)		
MOz A. saccharum	BAI	0.58	0.32	0.26		
	Tree Ring	0.58	0.23	0.35		
MOz Q. alba	BAI	0.83	0.41	0.42		
	Tree Ring	0.82	0.23	0.59		
MMS A. saccharum	BAI	0.72	0.24	0.48		
	Tree Ring	0.59	0.21	0.38		
MMS Q. alba	BAI	0.65	0.09	0.56		
	Tree Ring	0.63	0.06	0.57		
MMS L. tulipifera	BAI	0.63	0.25	0.38		
	Tree Ring	0.61	0.12	0.49		
UMB A. rubrum	BAI	0.18	-0.006	0.18		
	Tree Ring	0.49	0.21	0.28		
UMB Q. rubra	BAI	0.63	0.09	0.54		
	Tree Ring	0.66	0.22	0.44		
UMB P. grandidentata	BAI	0.51	0.02	0.49		
	Tree Ring	0.24	0.08	0.16		



Figure 5. Time-varying parameter regression slope estimates (black line) of the relationship between standardized tree growth and June, July, and August climatic water balance (P-PET), colored shading representing the 95% confidence interval. ACSA = Acer saccharum; QUAL = Quercus alba; LITU = Liriodendron tulipifera; ACRU = Acer rubrum; QURU = Quercus rubra; POGR = Populus grandidentata.

the influences among the variables was not consistent across the isotopic metrics (Δ^{13} C and iWUE) within a given species. Climatic water balance was a significant predictor in three species SEMs, having a very strong influence on *L. tulipifera* Δ^{13} C and iWUE at MMS and influencing *P. grandidentata* and *Q. rubra* Δ^{13} C at UMB (Figure 7 and supporting information Figures S10–S12). Both variables for pollution had limited influence on the isotopic metrics, with SO₄ and N deposition only impacting *A. rubrum* at UMB and *A. saccharum* at MOz, respectively (Figure 7 and supporting information Figures S10–S12). However, the inclusion of SO₄ improved model fit for another four species SEMs and N deposition improved three species SEMs (supporting information Figures S10–S12).

4. Discussion

4.1. Temporal Changes in Climate

While tree growth in the mesic eastern and Midwestern United States is sensitive to drought (Cook & Jacoby, 1977; Graumlich, 1993; Pederson et al., 2012, 2013; Rollinson et al., 2016; D'Orangeville et al., 2018; Canham



Figure 6. Time series of moving correlations of the early portion (May–June [MJ]; top six panels) and the later portion (August–September [AS]; bottom six panels) of the growing season P-PET with standardized tree rings (left column) and standardized basal area increment (right column). Correlations were done in 40-year moving windows with the year on the *x* axis representing the last year of the interval. ACSA = *Acer saccharum*; QUAL = *Quercus alba*; LITU = *Liriodendron tulipifera*; ACRU = *Acer rubrum*; QURU = *Quercus rubra*; POGR = *Populus grandidentata*.

et al., 2019), recent work showed a weakening relationship between radial growth and soil moisture starting in the early 1980s across sites and species in the Midwest (Maxwell et al., 2016). Our findings supported the recent weakening in response to water availability, with all *Quercus* species and *P. grandidentata* exhibiting a weakening relationship (Figures 4 and 5). However, by adding species not included in Maxwell et al. (2016) and by examining younger trees, we found that both *A. saccharum* and *A. rubrum* had stable relationships across our study region and *L. tulipifera* had become more sensitive to water availability (Figures 4 and 5). These results were well aligned with the relative degree of isohydricity of these species (i.e., *Quercus* are more anisohydric and *L. tulipifera* is very isohydric). Our findings suggested that all trees are sensitive to water availability when very dry conditions are included in the analysis. Even the anisohydric *Quercus* species responded to variations in P-PET when the drier 1950s were included. However, during periods where

100





Figure 7. Relative influence of climatic water balance, atmospheric CO_2 concentrations, and SO_4 and N wet deposition on the standardized tree rings and intrinsic water-use efficiency using fitted piecewise structural equation models for the period 1985–2015. Single-headed arrows represent causal relationship while double-headed arrows indicate covariance between variables. Solid line represents a coefficient that was significant in the SEM, while dotted lines represent coefficients that were not significant but improved model fit. Blue solid lines represent significant causal relationship, green dashed lines represent insignificant influence of variable but improved fit of the model, and gray lines represent the covariation of the explanatory variables. ACSA = Acer saccharum; QUAL = Quercus alba; LITU = Liriodendron tulipifera; ACRU = Acer rubrum; QURU = Quercus rubra; POGR = Populus grandidentata.

droughts are lacking or infrequent, radial growth of anisohydric *Quercus* can become insensitive to water availability compared to periods of drought. We posit that during wetter periods, *Quercus* are less sensitive to water availability compared to other species because they have relatively deeper roots (Abrams, 1990; Hinckley et al., 1981) and therefore have continuous access to water, allowing for a more aggressive water-use strategy. Conversely, species that have shallower roots, such as *Acer* and *L. tulipifera* (Abrams, 1990; Hinckley et al., 1981) will have inconsistent access to water during short periods of dryness within the general wet conditions and therefore employ a more conservative water-use strategy. Wood anatomy (i.e., ring porous versus diffuse porous) is another species trait that aligns with our results. We found that species with diffuse porous wood anatomy (e.g., *Acer* species and *L. tulipifera*), tended to remain more sensitive to water availability during the ongoing pluvial, an intuitive finding given recent evidence that indicates radial growth in diffuse porous species wood anatomy are more sensitive to drought (Elliott et al., 2015; Kannenberg et al., 2019; Yi et al., 2019). While wood anatomy is not directly aligned with water-use strategy, differences between diffuse and ring porous wood results in different xylem architecture and therefore are related to stomatal regulation (Elliott et al., 2015).

Generally, we found a more widespread and consistent weakening of the signal at sites that have not experienced drought recently and saw more stable relationships at the sites that have had recent drought (Figures 4 and 5). Interestingly, tree growth and carbon isotope metrics at UMB were not correlated strongly with climate variables including JJA P-PET for the period of 1970–2015 (supporting information Figures S7–S9), indicating the most recent decline in sensitivity to P-PET may not be representative of the long-term growth responses at this location. The relatively more stable correlations in the early portion of the growing season (MJ) compared to the later portion (AS), indicates that conditions during AS are more influential to the overall weakening in the JJA relationship (Figure 6). While the correlations are generally weaker than the JJA season, the changing correlations in AS season are important due to this being a relatively drier portion of the growing season. However, firm conclusions on how the conditions in AS are contributing to the JJA



relationship are difficult to draw due to some sites (e.g., MOz and MMS LITU) increasing in correlation, while species at other sites (UMB, MMS ACSA, and MMS QUAL) are decreasing (Figure 6).

4.2. Importance of Water Availability

Our results showed that while CO_2 concentrations, and to a lesser extent SO_4 and N deposition, influence the isotopic composition of tree rings, these variables resulted in a negligible influence on tree growth across species and sites over the timescale of our analysis. While increases in CO_2 concentrations have been shown to increase iWUE (Guerrieri et al., 2019), there is less support for growth enhancement from CO_2 due to lags between carbon allocation and stem growth and stand competition (Levesque et al., 2019; Giguere-Croteu et al., 2019). Further, the disagreement of the magnitude of the CO_2 fertilization effect on growth across data sources (e.g., Norby et al., 2005; Peñuelas et al., 2011, 2017) suggest that our understanding of how elevated CO_2 will impact tree growth and carbon sequestration when simultaneously being impacted by other pollutants and climate remains poorly understood. Our findings supported the work suggesting increases in atmospheric CO_2 influence iWUE but have weak if any influence on actual tree growth (Levesque et al., 2017; Peñuelas et al., 2011, 2017).

Considering that the period of analysis (1985–2015) is one of the wettest periods in the last several centuries (Ford, 2014; Maxwell & Harley, 2017), the continued importance of water availability on tree growth (while considering the simultaneous influence of CO_2 concentrations and SO_4 and N deposition) is particularly striking (Figure 7). CO_2 concentrations were more consistently related to the isotopic metrics than to the tree growth metrics (Figure 7), but that is partially explained by the inclusion of CO_2 in the calculation of iWUE (equation (3)). Regardless, the small influence of CO_2 concentrations on growth metrics indicates that increases in CO_2 are not contributing to the weakening relationship between tree rings and the climatic water balance. Similarly, the lack of influence of SO_4 and N deposition suggested that the decrease in pollutants is not contributing to a decrease in sensitivity of tree growth to water availability. Importantly, examining the variables simultaneously explains little of the variance (often <20%) in the growth and isotopic metrics at the site that has not experienced a drought recently (UMB). However, radial growth from both *Q. rubra* and *A. rubrum* had stronger correlations pre-1980 (Figures 4 and 5). These results indicate that while UMB is generally a wet site, trees have only recently become less sensitive to water availability and that decrease did not correspond to an increase in sensitivity in any other variable in the SEMs (Figure 7).

Finally, the inconsistent influence of climatic water balance on isotopic metrics across species (Figure 7) demonstrates the importance of understanding species-specific water-use strategies and how they relate to carbon uptake. The highly contrasting responses of *L. tulipifera* and *Q. alba* at MMS to climatic water balance is also supported by previous studies (Roman et al., 2015; Yi et al., 2019), which together suggest that the stomates of *L. tulipifera* are particularly sensitive to both changing water supply (P) and variable demand (PET), where *Q. alba* is largely insensitive to variation in water supply. If photosynthetic capacity is not directly affected by hydrologic stress, then stomatal closure in response to both reduced P and increased PET will tend to increase iWUE, since the relationship between photosynthesis and conductance is non-linear, and tends to saturate at relatively high conductance rate that characterize species growing in energy-limited environments (Yi et al., 2019). Our findings also provide further evidence of the usefulness of isotopic metrics to understand species-specific responses. It is important to note that while tree growth is one aspect of the forest carbon cycle, global change drivers could impact multiple aspects beyond just tree growth. As time series of flux towers and remote sensing products increase in length, we should be able to extend our hypotheses past tree rings and into other carbon cycle processes and quantify the extent to which the responses of tree rings to climate variability impact forest carbon uptake (Kannenberg et al., 2019).

5. Conclusions

We examined the simultaneous influence of climatic water balance (P-PET), CO_2 concentrations, and SO_4 and N depositions on tree growth and stable C isotopic composition from tree rings. We found that water availability was the most important driver of radial tree growth, although the SEMs at the site that has not experienced drought explained little variance. Generally, isohydric species were more sensitive to drought than anisohydric species and had a more stable and consistent relationship with water availability through time. CO_2 concentrations influenced the variability of $\Delta^{13}C$ and iWUE but not growth, suggesting



that the documented weakening in the relationship between tree growth and water availability is not due to increases in CO_2 concentrations. Similarly, both SO_4 and N deposition had little to no influence on tree growth, indicating that the recent pluvial period in the Midwest is likely causing the recent weakening relationship. With water availability being the dominant limitation of growth, a period absent of drought has resulted in trees being generally less sensitive to variation in the availability of water. However, the lack of influence of CO_2 concentrations and pollutant deposition on growth indicates that changes in these factors will unlikely offset growth reductions when drought returns in these regions. Therefore, future increases in the frequency and severity of drought will likely continue to be a limiting factor of both tree growth and carbon sequestration even with increased concentrations of CO_2 and lower levels of pollutants.

Acknowledgments

We would like to thank Brynn Taylor for laboratory assistance. This research was supported by funding to J. T. M. and K. A. N. from the USDA Agriculture and Food Research Initiative Grant 2017-67013-26191. Funding for tree ring data collection was provided to J. T. M. by the Indiana University Vice Provost for Research Faculty Research Program. The tree ring and stable C isotope chronologies from this paper is archived at the International Tree-Ring Data Bank (https://www.ncdc.noaa.gov/dataaccess/paleoclimatology-data/datasets/ tree-ring).

References

- Abrams, M. D. (1990). Adaptations and responses to drought in Quercus species of North America. *Tree Physiology*, 7(1-2-3-4), 227–238. https://doi.org/10.1093/treephys/7.1-2-3-4.227
- Ainsworth, E. A., & Long, S. P. (2005). What have we learned from 15 years of free-air CO2 enrichment (FACE)? A meta-analytic review of the responses of photosynthesis, canopy properties and plant production to rising CO₂. New Phytologist, 165(2), 351–371. https://doi.org/ 10.1111/j.1469-8137.2004.01224.x
- Allen, C. D., Breshears, D. D., & McDowell, N. G. (2015). On underestimation of global vulnerability to tree mortality and forest die-off from hotter drought in the Anthropocene. *Ecosphere*, 6(8), 1–55. https://doi.org/10.1890/ES15-00203.1
- Andersen, J., Hilberg, S., Kunkel, K., & Center, M. R. C. (2012). Historical climate and climate trends in the Midwestern USA. US National Climate Assessment Midwest Technical Input Report, 1-18.
- Battipaglia, G., Saurer, M., Cherubini, P., Siegwolf, R. T., & Cotrufo, M. F. (2009). Tree rings indicate different drought resistance of a native (Abies alba Mill.) and a nonnative (Picea abies (L.) Karst.) species co-occurring at a dry site in Southern Italy. Forest Ecology and Management, 257(3), 820–828. https://doi.org/10.1016/j.foreco.2008.10.015
- Beguería, S., Vicente-Serrano, S. M., Reig, F., & Latorre, B. (2014). Standardized precipitation evapotranspiration index (SPEI) revisited: Parameter fitting, evapotranspiration models, tools, datasets and drought monitoring. *International Journal of Climatology*, 34(10), 3001–3023. https://doi.org/10.1002/joc.3887
- Belmecheri, S., Maxwell, R. S., Taylor, A. H., Davis, K. J., Freeman, K. H., & Munger, W. J. (2014). Tree-ring δ¹³C tracks flux tower ecosystem productivity estimates in a NE temperate forest. *Environmental Research Letters*, 9(7), 074011. https://doi.org/10.1088/1748-9326/ 9/7/074011
- Bishop, D. A., Beier, C. M., Pederson, N., Lawrence, G. B., Stella, J. C., & Sullivan, T. J. (2015). Regional growth decline of sugar maple (Acer saccharum) and its potential causes. Ecosphere, 6(10), 1–14. https://doi.org/10.1890/ES15-00260.1
- Bréda, N., Huc, R., Granier, A., & Dreyer, E. (2006). Temperate forest trees and stands under severe drought: A review of ecophysiological responses, adaptation processes and long-term consequences. *Annals of Forest Science*, 63(6), 625–644. https://doi.org/10.1051/ forest:2006042
- Briffa, K. R., Jones, P. D., Bartholin, T. S., Eckstein, D., Schweingruber, F. H., Karlen, W., et al. (1992). Fennoscandian summers from AD 500: Temperature changes on short and long timescales. *Climate Dynamics*, 7(3), 111–119. https://doi.org/10.1007/BF00211153
- Brzostek, E. R., Dragoni, D., Schmid, H. P., Rahman, A. F., Sims, D., Wayson, C. A., et al. (2014). Chronic water stress reduces tree growth and the carbon sink of deciduous hardwood forests. *Global Change Biology*, 20(8), 2531–2539. https://doi.org/10.1111/ gcb.12528
- Bunn, A. G. (2008). A dendrochronology program library in R (dplR). Dendrochronologia, 26(2), 115–124. https://doi.org/10.1016/j. dendro.2008.01.002
- Canham, C. D., Murphy, L., Riemann, R., McCullough, R., & Burrill, E. (2018). Local differentiation in tree growth responses to climate. *Ecosphere*, 9(8), e02368. https://doi.org/10.1002/ecs2.2368
- Caspersen, J. P., Pacala, S. W., Jenkins, J. C., Hurtt, G. C., Moorcroft, P. R., & Birdsey, R. A. (2000). Contributions of land-use history to carbon accumulation in US forests. *Science*, 290(5494), 1148–1151. https://doi.org/10.1126/science.290.5494.1148
- Choat, B., Brodribb, T. J., Brodersen, C. R., Duursma, R. A., López, R., & Medlyn, B. E. (2018). Triggers of tree mortality under drought. *Nature*, 558(7711), 531–539. https://doi.org/10.1038/s41586-018-0240-x
- Choat, B., Jansen, S., Brodribb, T. J., Cochard, H., Delzon, S., Bhaskar, R., et al. (2012). Global convergence in the vulnerability of forests to drought. Nature, 491(7426), 752–755. https://doi.org/10.1038/nature11688
- Clark, J. S., Iverson, L., Woodall, C. W., Allen, C. D., Bell, D. M., Bragg, D. C., et al. (2016). The impacts of increasing drought on forest dynamics, structure, and biodiversity in the United States. *Global Change Biology*, 22(7), 2329–2352. https://doi.org/10.1111/gcb.13160
- Cook, E. R. (1981). The smoothing spline: A new approach to standardizing forest interior tree-ring width series for dendroclimatic studies. *Tree Ring Bulliten*, 41, 45–53.
- Cook, E. R. (1985). A time series analysis approach to tree ring standardization (Doctoral dissertation, University of Arizona).
- Cook, E. R. (1991). Tree rings as indicators of climate change and the potential response of forests to the greenhouse effect. In *Global climate change and life on earth*, (pp. 56–64). London, Routledge: Chapman & Hall.
- Cook, E. R., & Briffa, K. R. (1990). Data analysis. In E. R. Cook, & L. A. Kairiukstis (Eds.), Methods of Dendrochronology. Applications in the Environmental Sciences, (pp. 97–162). Kluwer Academic Publishers.
- Cook, E. R., & Jacoby, G. C. (1977). Tree-ring-drought relationships in the Hudson Valley, New York. Science, 198(4315), 399–401. https:// doi.org/10.1126/science.198.4315.399
- Cook, E. R., & Johnson, A. H. (1989). Climate change and forest decline: A review of the red spruce case. *Water, Air, and Soil Pollution,* 48(1-2), 127–140. https://doi.org/10.1007/BF00282374
- Cook, E. R., Johnson, A. H., & Blasing, T. J. (1987). Forest decline: Modeling the effect of climate in tree rings. *Tree Physiology*, 3(1), 27–40. https://doi.org/10.1093/treephys/3.1.27
- D'Orangeville, L., Maxwell, J., Kneeshaw, D., Pederson, N., Duchesne, L., Logan, T., et al. (2018). Drought timing and local climate determine the sensitivity of eastern temperate forests to drought. *Global Change Biology*, *24*(6), 2339–2351. https://doi.org/10.1111/gcb.14096



- Elliott, K. J., Miniat, C. F., Pederson, N., & Laseter, S. H. (2015). Forest tree growth response to hydroclimate variability in the southern Appalachians. *Global Change Biology*, *21*(12), 4627–4641. https://doi.org/10.1111/gcb.13045
- Engel, B. J., Schaberg, P. G., Hawley, G. J., Rayback, S. A., Pontius, J., Kosiba, A. M., & Miller, E. K. (2016). Assessing relationships between red spruce radial growth and pollution critical load exceedance values. *Forest Ecology and Management*, 359, 83–91. https://doi.org/ 10.1016/j.foreco.2015.09.029
- Farquhar, G. D., O'Leary, M. H., & Berry, J. A. (1982). On the relationship between carbon isotope discrimination and the intercellular carbon dioxide concentration in leaves. *Australian Journal of Plant Physiology*, 9, 121–137. https://doi.org/10.1071/PP9820121
- Farquhar, G. D., & Richards, R. A. (1984). Isotopic composition of plant carbon correlates with water-use efficiency of wheat genotypes. Functional Plant Biology, 11(6), 539–552. https://doi.org/10.1071/PP9840539
- Ford, T. W. (2014). Precipitation anomalies in eastern-Central Iowa from 1640-present. Journal of Hydrology, 519, 918–924. https://doi.org/ 10.1016/j.jhydrol.2014.08.021
- Gagen, M., McCarroll, D., Loader, N. J., & Robertson, I. (2011). Stable isotopes in dendroclimatology: Moving beyond "potential". In Dendroclimatology, (pp. 147–172). Dordrecht: Springer.
- Graumlich, L. J. (1991). Subalpine tree growth, climate, and increasing CO₂: An assessment of recent growth trends. *Ecology*, 72(1), 1–11. https://doi.org/10.2307/1938895
- Graumlich, L. J. (1993). Response of tree growth to climatic variation in the mixed conifer and deciduous forests of the upper Great Lakes region. *Canadian Journal of Forest Research*, 23(2), 133–143. https://doi.org/10.1139/x93-020
- Guerrieri, R., Belmecheri, S., Ollinger, S. V., Asbjornsen, H., Jennings, K., Xiao, J., et al. (2019). Disentangling the role of photosynthesis and stomatal conductance on rising forest water-use efficiency. *Proceedings of the National Academy of Sciences*, 116(34), 16909–16914.
- Hargreaves, G. H. (1994). Defining and using reference evapotranspiration. Journal of Irrigation and Drainage Engineering, 120(6), 1132–1139. https://doi.org/10.1061/(ASCE)0733-9437(1994)120:6(1132)

Hinckley, T., Teskey, R., Duhme, F., & Richter, H. (1981). Temperate hardwood forests. *Water deficit and plant growth*, *6*, 153–208. Holmes, R. L. (1983). Computer assisted quality control. *Tree-Ring Bulletin*, *43*, 69–78.

- Horn, K. J., Thomas, R. Q., Clark, C. M., Pardo, L. H., Fenn, M. E., Lawrence, G. B., et al. (2018). Growth and survival relationships of 71 tree species with nitrogen and sulfur deposition across the conterminous U.S. (ed Loustau D). PLoS ONE, 13(10), e0205296. https://doi. org/10.1371/journal.pone.0205296
- Hyvönen, R., Persson, T., Andersson, S., Olsson, B., Ågren, G. I., & Linder, S. (2008). Impact of long-term nitrogen addition on carbon stocks in trees and soils in northern Europe. *Biogeochemistry*, 89(1), 121–137. https://doi.org/10.1007/s10533-007-9121-3
- Johnson, A. H., Cook, E. R., & Siccama, T. G. (1988). Climate and red spruce growth and decline in the northern Appalachians. Proceedings of the National Academy of Sciences, 85(15), 5369–5373. https://doi.org/10.1073/pnas.85.15.5369
- Kalman, R. E. (1960). A new approach to linear filtering and prediction problems. Journal of Basic Engineering, 82(1), 35–45. https://doi. org/10.1115/1.3662552
- Kannenberg, S. A., Maxwell, J. T., Pederson, N., D'Orangeville, L., Ficklin, D. L., & Phillips, R. P. (2019). Drought legacies are dependent on water table depth, wood anatomy and drought timing across the eastern US. *Ecology Letters*, 22(1), 119–127. https://doi.org/10.1111/ ele.13173
- Kannenberg, S. A., Novick, K. A., Alexander, M. R., Maxwell, J. T., Moore, D. J., Phillips, R. P., & Anderegg, W. R. (2019). Linking drought legacy effects across scales: From leaves to tree rings to ecosystems. *Global Change Biology*, 25(9), 2978–2992. https://doi.org/10.1111/ gcb.14710
- Kannenberg, S. A., Novick, K. A., & Phillips, R. P. (2019). Anisohydric behavior linked to persistent hydraulic damage and delayed drought recovery across seven North American tree species. New Phytologist, 222(4), 1862–1872. https://doi.org/10.1111/nph.15699
- Karl, T. R., Knight, R. W., Easterling, D. R., & Quayle, R. G. (1996). Indices of climate change for the United States. Bulletin of the American Meteorological Society, 77(2), 279–292. https://doi.org/10.1175/1520-0477(1996)077<0279:IOCCFT>2.0.CO;2
- Keeling, C. D., Piper, S. C., Bacastow, R. B., Wahlen, M., Whorf, T. P., Heimann, M., & Meijer, H. A. (2005). Atmospheric CO₂ and ¹³CO₂ Exchange with the Terrestrial Biosphere and Oceans from 1978 to 2000: Observations and Carbon Cycle Implications. In I. T. Baldwin (Ed.), M. M. and Caldwell, and H. G., and J. R. B., and L. O.L., and M. H.A., ... and C. T. E (Eds.)*A history of atmospheric CO₂ and its effects on plants, animals, and ecosystems*, (pp. 83–113). New York, NY: Springer New York. doi: https://doi.org/10.1007/0-387-27048-5_5
- Kosiba, A. M., Schaberg, P. G., Rayback, S. A., & Hawley, G. J. (2018). The surprising recovery of red spruce growth shows links to decreased acid deposition and elevated temperature. *Science of the Total Environment*, 637-638, 1480–1491. https://doi.org/10.1016/j. scitotenv.2018.05.010
- LeBlanc, D. C., Raynal, D. J., & White, E. H. (1987). Acidic deposition and tree growth: II. Assessing the role of climate in recent growth declines. *Journal of Environmental Quality*, *16*(4), 334–340. https://doi.org/10.2134/jeq1987.00472425001600040008x
- Lefcheck, J. S. (2016). PIECEWISESEM: PIECEWISE STRUCTURAL EQUATION MODELLING IN R FOR ECOLOGY, EVOLUTION, AND SYSTEMATICS. *METHODS IN ECOLOGY AND EVOLUTION*, 7(5), 573–579. https://doi.org/10.1111/2041-210X.12512
- Levesque, M., Andreu-Hayles, L., & Pederson, N. (2017). Water availability drives gas exchange and growth of trees in northeastern US, not elevated CO₂ and reduced acid deposition. *Scientific Reports*, 7, 46158. https://doi.org/10.1038/srep46158
- Levesque, M., Andreu-Hayles, L., Smith, W. K., Williams, A. P., Hobi, M. L., Allred, B. W., & Pederson, N. (2019). Tree-ring isotopes capture interannual vegetation productivity dynamics at the biome scale. *Nature Communications*, 10(1), 742. https://doi.org/10.1038/s41467-019-08634-y
- Lévesque, M., Siegwolf, R., Saurer, M., Eilmann, B., & Rigling, A. (2014). Increased water-use efficiency does not lead to enhanced tree growth under xeric and mesic conditions. *New Phytologist*, 203(1), 94–109. https://doi.org/10.1111/nph.12772

Lipp, J., Trimborn, P., Fritz, P., Moser, H., Becker, B., & Frenzel, B. (1991). Stable isotopes in tree ring cellulose and climatic change. *Tellus B*, 43(3), 322–330. https://doi.org/10.1034/j.1600-0889.1991.t01-2-00005.x

- Livingston, N. J., & Spittlehouse, D. L. (1996). Carbon isotope fractionation in tree ring early and late wood in relation to intra-growing season water balance. *Plant, Cell & Environment*, 19(6), 768–774. https://doi.org/10.1111/j.1365-3040.1996.tb00413.x
- Loader, N. J., McCarroll, D., Gagen, M., Robertson, I., & Jalkanen, R. (2007). Extracting climatic information from stable isotopes in tree rings. *Terrestrial Ecology*, 1, 25–48. https://doi.org/10.1016/S1936-7961(07)01003-2
- Loader, N. J., Robertson, I., Barker, A. C., Switsur, V. R., & Waterhouse, J. S. (1997). An improved technique for the batch processing of small wholewood samples to α -cellulose. *Chemical Geology*, 136(3-4), 313–317. https://doi.org/10.1016/S0009-2541(96)00133-7
- Mathias, J. M., & Thomas, R. B. (2018). Disentangling the effects of acidic air pollution, atmospheric CO2, and climate change on recent growth of red spruce trees in the Central Appalachian Mountains. *Global Change Biology*, *24*(9), 3938–3953. https://doi.org/10.1111/gcb.14273

Maxwell, J. T. (2016). The benefit of including rarely-used species in dendroclimatic reconstructions: A case study using Juglans nigra in South-Central Indiana, USA. Tree-Ring Research, 72(1), 44–52. https://doi.org/10.3959/1536-1098-72.01.44

Maxwell, J. T., & Harley, G. L. (2017). Increased tree-ring network density reveals more precise estimations of sub-regional hydroclimate variability and climate dynamics in the Midwest, USA. *Climate Dynamics*, 49(4), 1479–1493. https://doi.org/10.1007/s00382-016-3396-9 Maxwell, J. T., Harley, G. L., & Matheus, T. J. (2015). Dendroclimatic reconstructions from multiple co-occurring species: A case study from

- an old-growth deciduous forest in Indiana, USA. International Journal of Climatology, 35(6), 860–870. https://doi.org/10.1002/joc.4021 Maxwell, J. T., Harley, G. L., & Robeson, S. M. (2016). On the declining relationship between tree growth and climate in the Midwest United
- States: The fading drought signal. *Climatic Change*, *138*(1-2), 127–142. https://doi.org/10.1007/s10584-016-1720-3 Maxwell, R. S., Hessl, A. E., Cook, E. R., & Pederson, N. (2011). A multispecies tree ring reconstruction of Potomac River streamflow (950–
- 2001). Water Resources Research, 47(5). https://doi.org/10.1029/2010WR010019
- McDowell, N. G., & Allen, C. D. (2015). Darcy's law predicts widespread forest mortality under climate warming. Nature Climate Change, 5(7), 669. https://doi.org/10.1038/nclimate2641
- McEwan, R. W., Dyer, J. M., & Pederson, N. (2011). Multiple interacting ecosystem drivers: Toward an encompassing hypothesis of oak forest dynamics across eastern North America. *Ecography*, 34(2), 244–256. https://doi.org/10.1111/j.1600-0587.2010.06390.x
- Mishra, V., Cherkauer, K. A., & Shukla, S. (2010). Assessment of drought due to historic climate variability and projected future climate change in the Midwestern United States. *Journal of Hydrometeorology*, *11*(1), 46–68. https://doi.org/10.1175/2009JHM1156.1
- Norby, R. J., DeLucia, E. H., Gielen, B., Calfapietra, C., Giardina, C. P., King, J. S., et al. (2005). Forest response to elevated CO₂ is conserved across a broad range of productivity. *Proceedings of the National Academy of Sciences USA*, 102, 18052–18056. https://doi.org/10.1073/ pnas.0509478102
- Norby, R. J., & Zak, D. R. (2011). Ecological lessons from free-air CO₂ enrichment (FACE) experiments. Annual Review of Ecology, Evolution, and Systematics, 42, 181–203. https://doi.org/10.1146/annurev-ecolsys-102209-144647
- Novick, K. A., Ficklin, D. L., Stoy, P. C., Williams, C. A., Bohrer, G., Oishi, A. C., et al. (2016). The increasing importance of atmospheric demand for ecosystem water and carbon fluxes. *Nature Climate Change*, 6(11), 1023. https://doi.org/10.1038/nclimate3114
- Ollinger, S. V., Aber, J. D., Reich, P. B., & Freuder, R. J. (2002). Interactive effects of nitrogen deposition, tropospheric ozone, elevated CO₂ and land use history on the carbon dynamics of northern hardwood forests. *Global Change Biology*, 8(6), 545–562. https://doi.org/ 10.1046/j.1365-2486.2002.00482.x
- Pederson, N., Bell, A. R., Cook, E. R., Lall, U., Devineni, N., Seager, R., et al. (2013). Is an epic pluvial masking the water insecurity of the greater New York City region? Journal of Climate, 26(4), 1339–1354. https://doi.org/10.1175/JCLI-D-11-00723.1
- Pederson, N., Bell, A. R., Knight, T. A., Leland, C., Malcomb, N., Anchukaitis, K. J., et al. (2012). A long-term perspective on a modern drought in the American Southeast. *Environmental Research Letters*, 7(1), 014034. https://doi.org/10.1088/1748-9326/7/1/014034
- Peñuelas, J., Canadell, J. G., & Ogaya, R. (2011). Increased water-use efficiency during the 20th century did not translate into enhanced tree growth. *Global Ecology and Biogeography*, 20(4), 597–608. https://doi.org/10.1111/j.1466-8238.2010.00608.x
- Peñuelas, J., Ciais, P., Canadell, J. G., Janssens, I. A., Fernández-Martínez, M., Carnicer, J., et al. (2017). Shifting from a fertilization-
- dominated to a warming-dominated period. Nature Ecology & Evolution, 1(10), 1438–1445. https://doi.org/10.1038/s41559-017-0274-8
 Peters, R. L., Groenendijk, P., Vlam, M., & Zuidema, P. A. (2015). Detecting long-term growth trends using tree rings: A critical evaluation of methods. Global Change Biology, 21(5), 2040–2054. https://doi.org/10.1111/gcb.12826
- Pettitt, A. N. A non-parametric approach to the change-point problem. Journal of the Royal Statistical Society, 28(2), 126-135. https://doi. org/10.2307/2346729
- R. Core Team. R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. https:// www.R-project.org/. (2017).
- Rollinson, C. R., Kaye, M. W., & Canham, C. D. (2016). Interspecific variation in growth responses to climate and competition of five eastern tree species. *Ecology*. https://doi.org/10.1890/15-1549
- Roman, D. T., Novick, K. A., Brzostek, E. R., Dragoni, D., Rahman, F., & Phillips, R. P. (2015). The role of isohydric and anisohydric species in determining ecosystem-scale response to severe drought. *Oecologia*, 179(3), 641–654. https://doi.org/10.1007/s00442-015-3380-9
- Schaberg, P. G., Hawley, G. J., Rayback, S. A., Halman, J. M., & Kosiba, A. M. (2014). Inconclusive evidence of Juniperus virginiana recovery following sulfur pollution reductions. Proceedings of the National Academy of Sciences, 111(1), E1–E1. https://doi.org/10.1073/ pnas.1320526111
- Schwede, D. B., & Lear, G. G. (2014). A novel hybrid approach for estimating total deposition in the United States. Atmospheric Environment, 92, 207–220. https://doi.org/10.1016/j.atmosenv.2014.04.008
- Soulé, P. T., & Knapp, P. A. (2006). Radial growth rate increases in naturally occurring ponderosa pine trees: A late-20th century CO₂ fertilization effect? *New Phytologist*, 171(2), 379–390. https://doi.org/10.1111/j.1469-8137.2006.01746
- Stahle, D. W., & Cleaveland, M. K. (1992). Reconstruction and analysis of spring rainfall over the southeastern US for the past 1000 years. Bulletin of the American Meteorological Society, 73(12), 1947–1961. https://doi.org/10.1175/1520-0477(1992)073<1947:RAAOSR>2.0. CO;2
- Thomas, R. Q., Bonan, G. B., & Goodale, C. L. (2013). Insights into mechanisms governing forest carbon response to nitrogen deposition: A model–data comparison using observed responses to nitrogen addition. *Biogeosciences*, 10(6), 3869–3887. https://doi.org/10.5194/bg-10-3869-2013
- Thomas, R. Q., Canham, C. D., Weathers, K. C., & Goodale, C. L. (2010). Increased tree carbon storage in response to nitrogen deposition in the US. *Nature Geoscience*, *3*(1), 13. https://doi.org/10.1038/ngeo721
- Vicente-Serrano, S. M., Beguería, S., López-Moreno, J. I., Angulo, M., & El Kenawy, A. (2010). A new global 0.5 gridded dataset (1901–2006) of a multiscalar drought index: Comparison with current drought index datasets based on the Palmer Drought Severity Index. *Journal of Hydrometeorology*, 11(4), 1033–1043. https://doi.org/10.1175/2010JHM1224.1
- Visser, H., Büntgen, U., D'Arrigo, R., & Petersen, A. C. (2010). Detecting instabilities in tree-ring proxy calibration. Climate of the Past, 6(3), 367–377. https://doi.org/10.5194/cp-6-367-2010
- Walker, A. P., De Kauwe, M. G., Medlyn, B. E., et al. (2019). Decadal biomass increment in early secondary succession woody ecosystems is increased by CO₂ enrichment. *Nature Communications*, 10(1), 454. https://doi.org/10.1038/s41467-019-08348-1
- Wigley, T. M., Briffa, K. R., & Jones, P. D. (1984). On the average value of correlated time series, with applications in dendroclimatology and hydrometeorology. *Journal of Climate and Applied Meteorology*, 23(2), 201–213. https://doi.org/10.1175/1520-0450(1984)023<0201: OTAVOC>2.0.CO;2
- Yamaguchi, D. K. (1991). A simple method for cross-dating increment cores from living trees. *Canadian Journal of Forest Research*, 21, 414–416. https://doi.org/10.1139/x91-053

IOC

- Yi, K., Maxwell, J. T., Wenzel, M. K., Roman, D. T., Sauer, P. E., Phillips, R. P., & Novick, K. A. (2019). Linking variation in intrinsic wateruse efficiency to isohydricity: A comparison at multiple spatiotemporal scales. *New Phytologist*, 221(1), 195–208. https://doi.org/10.1111/ nph.15384
- Zang, C., & Biondi, F. (2015). treeclim: An R package for the numerical calibration of proxy-climate relationships. *Ecography*, 38(4), 431–436. https://doi.org/10.1111/ecog.01335
- Zhu, Z., Piao, S., Myneni, R. B., Huang, M., Zeng, Z., Canadell, J. G., et al. (2016). Greening of the Earth and its drivers. *Nature Climate Change*, 6(8), 791. https://doi.org/10.1038/nclimate3004

Zuur, A. F., Ieno, E. N., Walker, N., Saveliev, A. A. & Smith, G. M. Mixed effects models and extensions in ecology with R.